CS190 Handins Handout #3 What you need to do with documents

• Prepare your documents with FrameMaker or Latex • Handin from your cs account (For the first three assignments, Requirements list, Specifications and Top level design, if you work in small group, handin one copy per group and be sure to indicate all the members) • Hand in a hardcopy to TA after the class on due day How to hand in documents from your cs account • Add /course/cs190/bin to your path • Make a subdirectory and put the file(s) you want to handin in it(If there are several files, be sure to include a short README to indicate that what each file is for) • cd into that directory • Type corresponding handin command(see below) Handin commands

• Requirements List(due on 2/3)

Type “cs190 handin req” • Specifications Document(due on 2/12)

Type “cs190 handin spec” • Top-Level Design(due on 2/24)

Type “cs190 handin topdsg” • Interface Proposals(due on 3/15)

Type “cs190 handin interface” • Detailed Design(due on 4/7)

Type “cs190 handin detaildsg”